Behav Ecol Sociobiol (2017) 71: 18 DOI 10.1007/s00265-016-2236-6

ORIGINAL ARTICLE

# Low incidence of inbreeding in a long-lived primate population isolated for 75 years

Anja Widdig<sup>1,2,3</sup> • Laura Muniz<sup>1,2</sup> • Mirjam Minkner<sup>1,2</sup> • Yvonne Barth<sup>1,2</sup> • Stefanie Bley<sup>1,2</sup> • Angelina Ruiz-Lambides<sup>1,2,4</sup> • Olaf Junge<sup>5</sup> • Roger Mundry<sup>6</sup> • Lars Kulik<sup>1,2</sup>

Received: 7 June 2016 / Revised: 9 November 2016 / Accepted: 11 November 2016 / Published online: 9 December 2016 © The Author(s) 2017. This article is published with open access at Springerlink.com

#### Abstract

When close relatives mate, offspring are expected to suffer fitness consequences due to inbreeding depression. Inbreeding has previously been quantified in two ways: using a sufficiently large panel of markers or deep and complete pedigrees over several generations. However, the application of both approaches is still limited by the challenge of compiling such data for species with long generation times, such as primates. Here, we assess inbreeding in rhesus macaques living on Cayo Santiago (Puerto Rico), a population genetically isolated since 1938, but descendant of

Communicated by E. Huchard

**Electronic supplementary material** The online version of this article (doi:10.1007/s00265-016-2236-6) contains supplementary material, which is available to authorized users.

Anja Widdig anja.widdig@eva.mpg.de

- <sup>1</sup> Junior Research Group of Primate Kin Selection, Department of Primatology, Max-Planck Institute for Evolutionary Anthropology, Deutscher Platz 6, 04103 Leipzig, Germany
- <sup>2</sup> Research Group of Behavioural Ecology, Institute of Biology, University of Leipzig, Talstraße 33, 04103 Leipzig, Germany
- <sup>3</sup> German Center for Integrative Biodiversity Research (iDiv), Deutscher Platz 5E, 04103 Leipzig, Germany
- <sup>4</sup> Caribbean Primate Research Center, University of Puerto Rico, Medical Sciences Campus, Punta Santiago, PO Box 906, San Juan, PR 00741, USA
- <sup>5</sup> Institute of Medical Informatics and Statistics, University Medical Center Schleswig-Holstein, Campus Kiel, Brunswiker Straße 10, 24105 Kiel, Germany
- <sup>6</sup> Max-Planck Institute for Evolutionary Anthropology, Deutscher Platz 6, 04103 Leipzig, Germany

a large set of presumably unrelated founders. Using comprehensive genetic data, we calculated inbreeding coefficients (F) for 2669 individuals with complete three generation pedigrees and 609 individuals with complete four generation pedigrees. We found that 0.79 and 7.39% of individuals had an F > 0 when using data from three and four generation pedigrees, respectively. No evidence of an increase in inbreeding over the study period (up to 23 years) was found. Furthermore, the observed mean relatedness of breeding pairs differed significantly from the distribution of parental relatedness expected as simulated based on previous reproductive data, suggesting that kin generally avoid breeding with each other. Finally, inbreeding was not a predictor of early mortality measured as survival until weaning and sexual maturation, respectively. Our results remain consistent with three estimators of inbreeding (standardized heterozygosity, internal relatedness, and homozygosity by loci) using up to 42 highly polymorphic microsatellites for the same set of individuals. Together, our results demonstrate that close inbreeding may not be prevalent even in populations isolated over long periods when mechanisms of inbreeding avoidance can operate.

# Significance statement

When close relatives mate, offspring may suffer from such inbreeding, e.g., via lower survival and/or fertility. Using (i) a large panel of genetic markers and (ii) complete three or four generation pedigrees, respectively, we show that incidences of inbreeding in a long-lived primate population are rare, even after genetic isolation for 75 years. Moreover, our simulations suggest that kin in our population generally avoid breeding with each other. Finally, the few inbred individuals detected in our large sample did not suffer from lower survival. Given that many animal species face dramatic habitat loss combined with critical population declines, our study provides important



implications for conservation biology in general and for population management in particular.

**Keywords** Multi-generational pedigree · Estimates of inbreeding · Inbreeding avoidance · Early mortality · Rhesus macaques · Genetic isolation

# Introduction

Individuals are considered as inbred when their parents are related to some degree. Studying inbreeding is important for our understanding of individual variation in fitness, as offspring of closely related parents are expected to have lower fitness than offspring of unrelated or distantly related parents, a phenomenon referred to as inbreeding depression (Charlesworth and Charlesworth 1987). Inbreeding increases "identity-bydescent" and therefore reduces heterozygosity of a given individual (Pemberton 2004; Slate et al. 2004; Taylor et al. 2010). Some of the deleterious effects observed in inbred offspring are lower chances of survival (Overall et al. 2005), lower fertility (Charpentier et al. 2005b), delayed development (Slate and Pemberton 2002), and a decreased immune function (Acevedo-Whitehouse et al. 2003; Rijks et al. 2008).

A wide range of inbreeding levels has been observed both in captivity and in the wild. For example, although some level of inbreeding has been reported in captive (Fredrickson and Hedrick 2002; Cassinello 2005; Ólafsdóttir and Kristjánsson 2008) and isolated populations (Marshall and Spalton 2000; Slate et al. 2000; Slate and Pemberton 2002), negligible levels were found in other populations of similar size (captive: Rzewuska et al. 2005; Dorostkar et al. 2012; Kanthaswamy et al. 2012; isolated: Overall et al. 2005). However, high to moderate levels of inbreeding have also been detected in wild populations (Liberg et al. 2005; Bensch et al. 2006; Prado-Martinez et al. 2013), with clear fitness costs at the individual level (Amos et al. 2001; Acevedo-Whitehouse et al. 2003; Bean et al. 2004; Rijks et al. 2008).

Several population characteristics are likely to lead to high levels of inbreeding over time, such as small founder population size, short-term population size reduction (bottleneck), lack of gene flow (isolation), or small population size (Marshall and Spalton 2000; Liberg et al. 2005; Bensch et al. 2006; Archie et al. 2007; reviewed in Hedrick and Kalinowski 2000; Keller and Waller 2002). Studies of inbreeding are important not only for ecology, evolution, and conservation biology, but also for population management, particularly of small, isolated, or fragmented populations being at risk of decreasing in heterozygosity (Marshall and Spalton 2000; Keller and Waller 2002).

Two alternative approaches have been used to estimate inbreeding: (i) calculating inbreeding coefficients from

pedigree data and (ii) measuring multi-locus heterozygosity from molecular markers. In the past, pedigrees were considered more accurate for estimating inbreeding (reviewed by Pemberton 2008), but even relatively few gaps influence the accuracy of estimates (Marshall et al. 2002). Particularly in wild and free-ranging populations, pedigree data tend to be incomplete and founders unknown (while assumed to be unrelated), which likely leads to inaccurate estimates of inbreeding (Keller and Waller 2002) and inbreeding depression (Taylor et al. 2015). However, all individuals of a species share common ancestry if traced back far enough in time; hence, most of their genome is identical by descent (Powell et al. 2010; Knief et al. 2015). Old coancestry is less problematic, as most recessive deleterious mutations exist at low allele frequencies and are likely to be lost over time. Large fitness effects are more likely a result of inbreeding between recent ancestors (Balloux et al. 2004; Pemberton 2004). Recent coancestry has been traditionally assessed by pedigrees, and simulation studies showed that with three generations of pedigree data, at least 80% of the variance in inbreeding is detected (Balloux et al. 2004). This is encouraging as construction of complete pedigrees with at least three generations should also be feasible for some wild or free-ranging mammalian populations.

When using genetic markers such as microsatellites (shorttandem repeats (STRs)), several studies revealed that multilocus heterozygosity is only weakly correlated with inbreeding coefficients derived from pedigree data (e.g., Balloux et al. 2004; Slate et al. 2004; Overall et al. 2005; Taylor et al. 2010). Consequently, when compared within the same study, inbreeding depression is often detected via pedigree, but not via multi-locus markers (e.g., Slate et al. 2004; Kim et al. 2007; but see Forstmeier et al. 2012). This appears to be more problematic when using a small panel of markers (typically 5 to 15) (Balloux et al. 2004) than when marker panels are large (e.g., Hoffman et al. 2014; but see Slate et al. 2004). Nevertheless, the exact number of loci required will depend on the extent to which genome-wide heterozygosity varies among individuals (Balloux et al. 2004), which is currently unknown for most populations (Hoffman et al. 2014).

Nowadays, marker-based estimates are expected to perform better than pedigrees when a large number of markers are used. Indeed, recent simulations suggested that inbreeding can be estimated more precisely with a large number of markers than with pedigree data (Kardos et al. 2015). However, empirical comparisons using a sufficiently large panel of markers in parallel to deep and complete pedigrees are still limited (e.g., Hoffman et al. 2014; Huisman et al. 2016), although combining the methods may contribute complementary information about population viability (Bensch et al. 2006). Compiling sufficiently deep and complete pedigrees together with at least a medium number of STR markers is particularly challenging for species with long generation times, such as primates, as it requires intense sampling over decades (cf. Szulkin et al. 2007). Nevertheless, long-term studies that allow the combination of both approaches are valuable for further advances in the study of inbreeding.

The aim of the present study was therefore to investigate inbreeding in a primate population isolated for more than seven decades. To do so, we applied both genetic markers and pedigree data to estimate inbreeding in a free-ranging population of rhesus macaques (Macaca mulatta) on the island of Cayo Santiago (Puerto Rico, USA) that lives in several naturally formed groups. The monkeys stem exclusively from 409 wild born animals captured in 12 districts in the Lucknow area of northern India and were transferred to Cayo Santiago in 1938 (details in Carpenter and Krakower 1941; Altmann 1962; Carpenter 1972; Rawlins and Kessler 1986). Since then, no animal has been added to the island except through births. Annual reproduction among male primates is typically skewed, i.e., restricted to a few males (reviewed in Widdig 2013), which was also reported for the study population based on short-term (Widdig et al. 2004; Dubuc et al. 2011) and lifetime reproduction (Dubuc et al. 2014). Hence, a potential founder effect and the continued isolation in combination with male reproductive skew might have decreased heterozygosity and increased the chance of individuals breeding with relatives over time, leading to inbreeding depression in this population.

We had three major goals with this study. First, we aimed to evaluate the proportion of offspring produced by closely related parents and whether it has increased over time. Second, we investigated whether inbreeding is avoided in our study population by comparing the observed relatedness of actual breeding pairs with the average relatedness as expected based on a simulation considering male reproductive skew, extra-group paternities, and natal breeding as revealed by previous studies. Finally, as inbreeding has previously been shown to reduce survival (e.g., Overall et al. 2005), we tested whether inbreeding is a predictor of early mortality by investigating different life history stages. To reach our aims, inbreeding coefficients were calculated for individuals with complete pedigree data over three or four generations. Such detailed pedigree data are still extremely limited for any free-ranging primate population. To complement our pedigree data, we conducted the same analyses with three estimates of inbreeding based on up to 42 STR markers available for this population.

# Methods

# Study species

Rhesus macaques live in multi-male, multi-female groups of 8 to 180 individuals in the wild (Seth and Seth 1986). Females are philopatric and form stable matrilineal hierarchies (Gouzoules and Gouzoules 1987), while males disperse from

their natal group (Lindburg 1969; Colvin 1983) between 3 to 5.5 years of age (median age = 4.5 years) (Berard 1990). Data from captivity suggest that female rhesus macaques reach sexual maturation between 2.5 and 3.5 years of age (Zehr et al. 2005) and males between 3 and 3.5 years of age (Dixson and Nevison 1997). Females usually give birth to a single off-spring per season (Rawlins and Kessler 1986). Offspring can be assigned to non-overlapping birth cohorts, although cohort members may differ in age by up to 6 months. Both males and females mate with several partners during the breeding season (Hoffman et al. 2008).

## The study population and its founding

Cavo Santiago is a 15.2-ha island off the coast of Puerto Rico (18° 09' N, 65° 44' W) managed by the Caribbean Primate Research Center (CPRC) (Rawlins and Kessler 1986). The population stems entirely from 409 wild born animals, including 183 adult females and 40 adult males, collected from 12 districts in the Lucknow area (comprising 2500 km<sup>2</sup>) in the central Uttar Pradesh state of northeastern India, which were released on the island in December 1938 (Rawlins and Kessler 1986). No animals have been added since, so the population has grown exclusively through natural births. However, although a considerable proportion of infants does not survive their first year of life (14.4% of infants born between 1992 and 2014, see Supplement) while the population endures natural disasters (hurricanes and disease outbreaks), removals of animals have been necessary to control the population size (see Hernández-Pacheco et al. 2013 for details on the culling strategies and the Supplement for proportions removed). Briefly, since 1985, either entire social groups were culled or 2-yearold males and females were randomly removed to achieve an adult sex ratio of two females per male following what is observed in wild populations (Hernández-Pacheco et al. 2013). Today, the population size is maintained at around 1000 individuals living in naturally formed social groups (mean  $\pm$  SD = 6.91  $\pm$  2.94 groups during our study period) after the most intense short-term bottleneck which reduced the population to around 300 individuals by the end of 1972 (Fig. 1, for details, see Widdig et al. 2016). Animals are provisioned with a commercial monkey diet (0.23 kg/monkey/ day) but spend at least 50% of their feeding time foraging extensively on natural vegetation (Marriott et al. 1989).

Data from previous studies using different methods, fewer animals, and no pedigree data suggested that the Cayo Santiago population is not inbred. A first study based on eight blood group loci sampled in 1972 reported an average heterozygosity of 0.324, indicating that a large amount of blood group variation remained (Duggleby et al. 1986). Although imprecise as a mean to measure inbreeding, a later study, assessing dyadic relatedness from multi-locus genotypes, revealed that 75% of all dyads investigated (excluding parent-offspring, full- or half-



Fig. 1 Colony growth from 1956 to 2011 showing the total number of animals recorded in the annual census

siblings) had a coefficient of relatedness of less than 0.0625 (Widdig et al. 2001). However, a more comprehensive assessment of current levels of inbreeding is now possible based on long-term genetic data (see Supplement).

Lifetime data on reproductive success suggest that males on Cayo Santiago reaching sexual maturation sire 8.7 offspring, on average (range 0-47; non-breeders 17.4% of males), while females give birth to 7.7 offspring, on average (range 0-16; non-breeders 4.5% of females) (Dubuc et al. 2014).

## **Demographic information**

Since 1956, trained CPRC staff has noted for all animals the ID, sex, date of birth, behavioral mother, birth group, and date of death or removal. When an individual disperses, daily census takers record its new group and check this assignment regularly for at least 2 months. If group membership remains constant, the first day seen in the new group is defined as the date of immigration for a given individual. The demographic dataset considered here comprises 11,715 animals that lived on Cayo Santiago between January 1938 and January 2011.

#### Genetic data for parentage analysis

Predominantly blood samples were collected continuously from the entire population with extensive sampling efforts since 1992 in order to perform paternity analysis (see Supplement for details). Genetic sampling of individuals is routinely conducted by the CPRC during annual trapping, mainly at 1 year of age; however, for a subset of subjects born in one group (troop R) between 2005 and 2014, we maximized efforts to sample entire birth cohorts (see "Statistical analyses"). Two independent genetic databases covering different sample sets and periods as well as different sets of markers are available today. Here, we combined both genetic datasets to maximize the genetic information available (see Supplement for details).

To date, genetic information is available for 4641 animals, genotyped on an average of  $27.6 \pm 1.6$  STR markers (mean  $\pm$ SD) (see Supplement for more details). We used highly polymorphic markers showing similar characteristics with regard to the number of alleles and heterozygosity as in several wild rhesus populations (Satkoski et al. 2008; Li et al. 2013). Paternity has been determined for 3934 individuals out of the 4014 individuals (98.0%) sampled between 1992 and 2014, and maternity, as derived from behavioral observations, could be confirmed genetically for 3946 of 3996 motheroffspring pairs (98.7%). Paternity was determined using a combination of exclusion and likelihood analyses by considering all sampled potential sires present on the island around conception of the respective infant (see Supplement for details). All cases of paternity were confirmed at the 95% level (N = 3931) or 80% level (N = 3) in CERVUS 3.0 (Kalinowski et al. 2007). While it was not possible to collect genetic samples in a blind fashion, people conducting the genetic analysis were uninformed about the study question.

## Calculation of inbreeding based on pedigree

The inbreeding coefficient (F) of an individual was calculated in R (version 3.2.1) (R Core Team 2015) using the package "pedigree" (version 1.4) (Coster 2012). Following previous studies (e.g., Slate et al. 2004), we assumed that F among all founders was close to zero, which seems quite likely, given that relatedness among individuals taken from different locations spread over a large area is likely to be low or zero. The inbreeding coefficient of an offspring is equal to one half of the coefficient of relatedness (r) between the parents of that offspring; e.g.,  $F_{\text{offspring}} = 0.25$  corresponds to  $r_{\text{parents}} = 0.5$ . The use of complete and deep pedigrees (i.e., three or more generations) was shown to provide a suitable approximation of the actual F (Balloux et al. 2004). Therefore, we calculated F(i) for the subset of animals with complete four generation pedigrees (N = 609 individuals born in the birth seasons between 2004 and 2014) and, to increase our sample size, (ii) for animals with complete three generation pedigrees (N = 2669individuals born in the birth seasons between 1992 and 2014). Both datasets included the majority of groups present during the study period and encompassed 11 and 23 study years (or birth cohorts), respectively.

## Estimating inbreeding based on genetic markers

Various marker-based estimators have been used to investigate the genome-wide diversity of populations (Pemberton 2008; Wang 2011). We calculated three commonly used estimators of inbreeding: standardized heterozygosity (SH), internal relatedness (IR), and homozygosity by loci (HL) (Slate et al. 2004; Charpentier et al. 2005b; Overall et al. 2005; Ruiz-López et al. 2009) to evaluate their performance in comparison to *F*. To achieve comparable data, the same set of animals was used as described previously. For animals with complete four generation pedigrees (N = 609), we had  $28.4 \pm 3.7$  (mean  $\pm$  SD) with a range of 13–41 markers available, which was similar to the subset of animals with complete three generation pedigrees (N = 2669) (mean  $\pm$  SD =  $31.0 \pm 5.6$ , range markers = 11-42).

SH calculates the proportion of heterozygous loci divided by the mean heterozygosity of all loci typed per individual (Coltman et al. 1999). This inbreeding estimate can attain values from 0 to infinite, the lower the value, the more loci are homozygous (indicating an inbred individual). IR measures the homozygosity through counting the frequencies of shared alleles between parents giving rare shared alleles more weight than common shared alleles (Amos et al. 2001). This estimator ranges from -1 to 1, where negative values indicate offspring of more outbred parents and positive values offspring of more inbred parents. A solution for the possible underestimation of heterozygosity in animals carrying rare alleles is HL (Aparicio et al. 2006). This metric for assessing genetic variability weights heterozygosity by the diversity of each homozygous locus (Aparicio et al. 2006; Rijks et al. 2008). Loci with higher allelic variability are assumed to be more informative and thus are weighted more than others. HL ranges from 0 to 1 with 0 indicating that all loci are heterozygous and 1 suggesting that all loci are homozygous. Based on the definitions of these estimators, values of HL and SH as well as values of IR and SH were negatively correlated, while values of HL and IR were positively correlated (see Supplement). All estimators of inbreeding were calculated in R (version 2.15.1) (R Development Core Team 2010) using the package "Rhh" (Alho et al. 2010).

## Statistical analyses

#### Evaluation of inbreeding over time

We used a generalized linear mixed model (GLMM) (Baayen 2008) to assess the proportion of offspring produced by closely related parents across time in the study population. The full model comprised consecutively numbered birth cohorts ("birth season") as a fixed effect and the IDs of the birth group, the dam, and the sire as random effects. Since it has been shown that neglecting random slopes can lead to drastically inflated type I error rates (i.e., false significance; Schielzeth and Forstmeier 2009; Barr et al. 2013), we included random slopes of birth season within sire and group but not within dam since for too many dams, we had only one birth in the data (making the random slope nearly "unidentifiable"; Barr et al. 2013). On a biological level, such random slopes account for the possibility that, for instance, the change of inbreeding over time varies among groups (a possibility that seems likely, given that groups vary considerably with regard to their size or number and size of neighboring groups) or sires (an option conceivable, too, since males might be of different attractiveness to females). Initially, we used F as a binary response (yes for F > 0, no for F = 0), which revealed "false convergence." We thus transformed the F values into ranks, where the smallest value, F = 0, was set to zero, F = 0.0625 to 1, F = 0.125 to 2, and F = 0.25 to 3 and used a Poisson model. We fitted two such models, one based on all animals for which F could be calculated for three complete generations and one included all animals with four complete generations. The model was implemented in R (version 3.2.1) (R Core Team 2015) using function "glmer" of the R package "lme4" (version 1.0–8) (Bates et al. 2015). The P values for birth cohort were determined using a likelihood ratio test (Barr et al. 2013). Overdispersion was not an issue (three generations:  $\chi^2 = 1.567, df = 2667, P = 1$ , dispersion parameter = 0.001; four generations:  $\chi^2 = 56.867$ , df = 607, P = 1; dispersion parameter = 0.0934); however, there was indication of underdispersion, presumably reflecting the small number of inbreeding cases observed in our dataset.

In three additional pairs of models, we used the same approach, but F was replaced by one of the three inbreeding estimators SH, IR, and HL, respectively. Due to the normal distribution of the response, we here applied a Gaussian error distribution, i.e., used a linear mixed model (LMM), fitted using the function "lmer." To assess the model stability, we compared the estimates derived by a model based on all data with those obtained from models with the levels of the random effects excluded one at a time. These revealed that the models were stable.

Finally, we calculated for each model the marginal  $R^2$  ("variance explained"), which was suggested as an appropriate measure for the goodness of fit of a model (Nakagawa and Schielzeth 2013).

#### Evaluation of inbreeding avoidance on Cayo Santiago

To investigate whether relatives avoid breeding in our study population, we compared the observed average relatedness of breeding pairs revealed from pedigree data with the average relatedness expected based on a simulation considering previous reproductive data using a permutation test (Adams and Anthony 1996; Manly 1997). Following previous studies (e.g., Keller and Arcese 1998; Szulkin et al. 2009; Rioux-Paquette et al. 2010), we ran different simulations of the natural scenario based on published data (details in the following). To do so, we calculated relatedness from pedigrees using an R script (available upon request from the authors) for all potential breeding pairs formed by each of the available mothers of the offspring born after 1996 with a randomly chosen potential sire (chosen from all males alive and mature at the time that each of these offspring was conceived).

Several studies on Cayo Santiago revealed that male reproductive success is moderately skewed compared to other species. For example, genetic paternity success in six consecutive birth cohorts of one large group (troop R) revealed that the most successful male sired on average 24% of offspring (Widdig et al. 2004) with an average of 29.3% of group males reproducing across years (AW, unpublished data). As a high male reproductive skew could increase the likelihood of inbreeding (if age cohorts are, e.g., sired by one or two males only) and to rule out the possibility that selective culling might have reduced the likelihood of inbreeding, we used a higher skew than observed in the study population for our simulation. The same study showed that 40.9% of all infants were sired by group males of an age between 9 and 11 years (Widdig et al. 2004). Hence, we randomly chose only 20% of the males with an age between 9 and 11 years per group as potential sires to contribute to a given birth cohort per group in order to simulate male reproductive skew. Furthermore, a previous study in group R found that, on average, 24% of offspring were assigned to sires belonging to different social groups (hereafter extra-group paternities) (Widdig et al. 2004). Hence, in our simulation, 24% of offspring were randomly assigned to potential sires outside of their natal group. Finally, as 16% of offspring born in the population were sired by natal males prior to dispersal (AW, unpublished data), we consequently also included natal males as potential sires in our simulation. In summary, in each simulation, 24% of offspring were randomly assigned to extra-group males and 16% of offspring to natal males and the remaining 60% of offspring were assigned to non-natal group males of which we considered only 20% of the available group males to account for reproductive skew (i.e., simulating no natal dispersal). To investigate whether the proportion of natal males could affect the average relatedness occurrence, we ran a second simulation using the same approach as described previously, but excluding natal males as potential breeders (i.e., simulating natal dispersal of all males).

To minimize the bias due to incomplete pedigree data, we used only potential (and actual) breeding pairs with complete pedigrees for at least three generations. This simulation included 966 offspring (born after 1996), their respective mothers (N = 379), and all of their potential sires (N = 620). We repeated each simulation 1000 times to generate 1000 different datasets (into which we included the observed data as one dataset) where mothers were randomly paired with a potential sire in order to obtain the parental relatedness expected based on a simulation considering previous reproductive data. We finally determined the P value as the proportion of simulations revealing an average relatedness of parents at least as small as the actually observed value. If this reveals significance, it means that inbreeding was avoided.

## Testing inbreeding as a cause of early mortality

We considered two different life history stages in this analysis (survival until 1 year of age and up to sexual maturation). First, we tested whether inbreeding was a possible cause of early infant mortality (survival until 1 year of age, i.e., weaning age). To this end, we calculated F for a subset of infants of our study population. Given that infant mortality in our population is highest within the first year of life (Blomquist 2013), genetic sampling at 1 year of age in general was never complete for an entire birth cohort (see Supplement). In addition, primate mothers carry dead infants for several days (e.g., Sugiyama et al. 2009), making sampling of dead infants extremely challenging under free-ranging conditions. To investigate inbreeding as a potential cause of mortality, we used data from one group (troop R) by maximizing sampling efforts for all 610 infants born between 2005 and 2014. For this subset, we either aimed at collecting the following: (i) hair samples as soon as possible after birth (1–5 months of age) during trapping, (ii) fecal samples if infants were considered sick or weak, or (iii) hair or tissue samples if infants were found dead (i.e., after the mother gave up carrying the infant). In total, we were able to sample 533 of the 610 live births (87.4%). To evaluate our sampling effort, we compared birth cohorts (or years) with a high sampling rate (>90%, mean  $\pm$  SD 92.92  $\pm$  0.03) with years of lower sampling rate (<90%, mean  $\pm$  SD 77.88  $\pm$  0.11) with regard to the mean observed inbreeding (measured as parental relatedness). Subsequently, we calculated 95% confidence intervals and found them clearly overlapping (years with sampling rates >90%, 0.000 to 0.005; years <90%, 0.002 to 0.007) indicating no obvious differences between the amounts of inbreeding detected in years of different sampling efforts.

For 516 of these infants, we were able to determine complete pedigrees for three generations, which revealed only three cases of inbreeding in this subset (0.58%). However, to reveal more distant events of inbreeding with a lower F value (Balloux et al. 2004), we incorporated the fourth generation by accepting incomplete pedigrees of all 516 infants, too, leading to a total of 38 inbred infants (7.36%) (F range 0.0313–0.25) in this subset. Of the 516 infants considered, 40 died (7.75%), while 15 were removed (2.91%) from the population before completing the first year of life. In our analysis on early infant mortality, we did not include removed subjects, leading to a total number of 501 individuals. However, this did not bias our sample as the mean F values were higher for infants not removed (0.004) (including dead infants) than for removed individuals (0.000).

Previous studies indicated that effects of inbreeding depression on mortality might increase until maturation (e.g., Ralls et al. 1988). Therefore, evaluating inbreeding depression based on a single (and early) life history stage might underestimate the costs of inbreeding (Szulkin et al. 2007; Grueber et al. 2010). Following a previous study (Bilski et al. 2013), in addition to survival until 1 year of age, we also looked at survival to the time of sexual maturation. Given the age of maturation reported in a previous study (see above), we assumed that by the age of 4 years both sexes have reached sexual maturation. Of the 516 individuals born, 77 died (14.92%), while 160 individuals were removed (31.01%) before completing the fourth year of life, but again, this did not create a bias of our sample as the mean F value was higher for individuals not removed (0.004) than for individuals who were removed (0.003). Among the removed subjects, there were a few inbred animals, which reduced the total number of inbred individuals from 38 (see above) to 27 individuals (7.58%) in this subset (F range 0.0313-025). Of the 516 offspring born, we did not consider the 160 subjects removed before maturation, which resulted in a total number of 356 subjects included in this analysis. Note that due to the long life span in this species, our pedigree data were not sufficient to consider the effect of inbreeding on lifetime reproductive success.

We applied a mixed Cox regression using the function "coxme" provided by the R package coxme (Therneau 2015) in order to investigate whether inbreeding caused mortality before 1 or 4 years of age, respectively. As the Cox regression requires, we used as response the number of surviving days and the appropriate status (dead/alive) at the end of the period considered (one or 4 years of age, respectively, for individuals that survived or the age at which the infant died). Hence, animals surviving at least 1 or 4 years of their lives were assigned as being 365 or 1460 days old, respectively. Per response (survival until 1 year of age or maturation), we fitted four separate Cox models including as predictor either F (calculated up to four generations deep) or the three estimators of inbreeding (SH, IR, HL), respectively. As random effects, we included the identity of the mother and the father (as several of them occurred repeatedly in the dataset) in all models. As our dataset included very few cases of inbreeding, we additionally used GLMMs with binomial error structure and logit function. As response, we used a variable indicating whether or not a given infant survived at least 1 or 4 years of their lives, respectively. As fixed effects and random effects, we used the same set as described previously. The models were implemented in R (version 3.2.1) (R Core Team 2015) using function glmer of the R package lme4 (version 1.0-8) (Bates et al. 2015). The P values for the fixed effects were determined using a likelihood ratio test (Barr et al. 2013).

# Results

## **General results**

Using complete pedigrees encompassing three generations, we found an F > 0 for 21 out of 2669 individuals investigated (0.79%) with 0.43 to 2.03% of inbred offspring

born in 12 out of 23 birth cohorts considered. However, looking at four complete generations of pedigrees, we found 45 out of 609 individuals (7.39%) with F > 0 including 1.82 to 8.41% of offspring born in all 11 birth cohorts considered. As theoretically expected (Balloux et al. 2004), F values decreased, while the proportion of inbred offspring increased from three to four generation pedigrees (compare Tables 1 and 2).

The 21 inbred offspring detected within the three generation pedigrees (i.e., up to the grandparent) (mean F = 0.14) were almost exclusively inbred within the paternal line. Specifically, we detected a total of 18 cases of inbreeding between paternal half-siblings (in 11 of these, the sire was still in his natal group and often high ranking, whereas all the remaining cases were the result of extragroup paternity), and in two cases, inbreeding concerned father-daughter pairs (with one case of an extra-group paternity). The remaining case concerned inbreeding between maternal half-siblings who were never coresident in the same group (i.e., the brother left the group before the sister was born) with the inbred offspring resulting from extragroup paternity (Table 1).

Looking at the 45 inbred offspring within four generation pedigrees (i.e., up to great grandparent, hence inbreeding deeper in the pedigree was revealed) (mean F = 0.04), the highest *F* was found again among paternal kin (i.e., paternal half-siblings); however, the majority of parents producing inbred offspring were related via both the maternal and the paternal lines (Table 2, see Fig. S1a–c in Supplement).

#### Evaluation of inbreeding over time

To assess inbreeding over time, we fitted a GLMM or LMM, separately for each inbreeding measure including individuals of consecutive birth cohorts. Regardless of the inbreeding measure used as response, the results revealed that the level of inbreeding did not change significantly over time with the results of the model being consistent for three and four generation pedigree data (Tables 3 and 4). Over all models, the marginal  $R^2$  values ranged between 0.001 and 0.003, which supports our finding that variance of inbreeding was not well explained by birth season.

#### Evaluation of inbreeding avoidance on Cayo Santiago

Our observed relatedness of actual breeding pairs showed that 92.4% of pairs were unrelated (Table 5). Although the distribution was skewed towards zero due to the majority of unrelated breeders, it was best represented by its mean ( $r = 0.008 \pm 0.033$ , mean  $\pm$  SD, range 0.0–0.5) in comparison to other distribution measures. Therefore, we also used the mean to compare the observed and simulated populations.

**Table 1**Inbreeding detected inthe three-generation pedigree data

F	Parent generation with inbreeding noted between	Kin line	N cases
0.25	Daughter/sire	Paternal	2
0.125	Paternal half-sibs	Paternal	18
0.125	Maternal half-sibs	Maternal	1

Note that cases of inbreeding detected in three-generation pedigrees were only considered in the four-generation analysis if the four-generation pedigree was complete. Likewise, cases detected only in the four-generation pedigree (e.g., the common ancestor is a great grandparent) remain undetected when considering only three generations

The observed mean relatedness of pairs that reproduced in the population (arrow in Fig. 2) was clearly below the distribution of parental relatedness expected while considering reproductive skew, extra-group paternity, and natal breeders (gray bars in Fig. 2, permutation test, P = 0.001), suggesting inbreeding avoidance. Over this set of simulations (allowing natal breeding), the relatedness values had a mean of  $0.012 \pm 0.002$  (mean  $\pm$  SD) and ranged between 0.008 and 0.017, while the proportion of values >0 within each simulation was  $0.093 \pm 0.009$  (mean  $\pm$  SD). To further investigate the importance of natal breeders on the occurrence of inbreeding, we ran the same simulation again, this time excluding the possibility of natal breeding (i.e., simulating natal dispersal of all males). This led to smaller mean relatedness values of potential breeders (hatched bars in Fig. 2) and a nonsignificant difference between the observed and expected mean relatedness (permutation test, P = 0.204). Over this set of simulations (excluding natal breeding), the relatedness values had a mean of  $0.009 \pm 0.001$  (mean  $\pm$  SD) and ranged between 0.005 and 0.014, while the proportion of values >0within each simulation was  $0.073 \pm 0.008$  (mean  $\pm$  SD). Comparing both distributions, this illustrates that natal breeding indeed increases the chance of inbreeding. Overall, our results reveal that kin breed less often than expected based on a simulation considering male reproductive skew, extragroup paternities, and natal breeding, suggesting avoidance of breeding with kin.

# Testing inbreeding as a cause of early mortality

The Cox regression using F for three generations as a predictor did not converge, as expected, given that only two inbreeding cases were detected. Using F calculated for up to four, partly incomplete generations, the results of the Cox regression did not provide any evidence of inbreeding affecting the probability of early infant death (Table 6). Similarly, no effect of inbreeding was found on the probability to survive until maturation (Table 7). The results did not change when using the three other inbreeding estimates (Tables 6 and 7) or when applying binomial regression analyses on these data (Tables S6 and S7 in Supplement).

# Discussion

This study assessed the incidence of inbreeding based on deep and complete pedigree data in parallel to a large panel of genetic markers in a long-lived primate species. Overall, the results indicate that, even after more than seven decades of genetic isolation, events of inbreeding are rare in the rhesus macaque population of the Cayo Santiago island. First, we found no evidence of an increase in inbreeding over up to 23 years, regardless of the inbreeding measure used. Second, the observed mean relatedness of pairs that had reproduced in the population was clearly below the distribution expected

F	Parent generation with inbreeding noted between	Kin line	N cases
0.125	Paternal half-sibs	Paternal	4
0.03125	Cousins	Maternal	1
0.03125	Cousins	Paternal	4
0.03125	Cousins	Maternal and paternal	29
0.0625	Uncle/niece	Maternal and paternal	1
0.0625	Aunt/nephew	Maternal and paternal	3
0.0625	Uncle/niece	Paternal	2
0.0625	Aunt/nephew	Paternal	1

Same note as in Table 1

 Table 2
 Inbreeding detected in

 the four-generation pedigree data

	Est	SE	$\chi^2$	df	P values	2.5% CL	97.5% CL
F							
Intercept	-15.986	2.563				-21.169	-15.501
Birth season	-0.198	1.983	0.060	1	0.806	-1.151	0.220
IR							
Intercept	-0.001	0.002				-0.005	0.005
Birth season	0.001	0.002	0.062	1	0.804	-0.003	0.005
SH							
Intercept	0.001	0.003				-0.004	0.007
Birth season	-0.001	0.003	0.081	1	0.776	-0.006	0.003
HL							
Intercept	-0.001	0.002				-0.004	0.003
Birth season	0.001	0.002	0.622	1	0.430	-0.003	0.005

Table 3Results of the GLMM and LMMs testing inbreeding over time assessed by F and inbreeding estimates (IR, SH, and HL) using the samesubset of 2669 individuals (three complete generation pedigrees)

based on a simulation, suggesting that breeding with kin occurred less often than expected by chance and may be avoided. Finally, when investigating whether mortality until weaning or maturation (1 or 4 years of age, respectively) was influenced by inbreeding, we found no obvious effects for either life history stage. Overall, this suggests that the low degree of inbreeding found in the study population did not result in fitness costs as measured by offspring survival.

Several mechanisms for inbreeding avoidance, such as sexbiased dispersal, extra-group paternity, and kin recognition, have been suggested and reported in a wide range of mammalian species (Kuester et al. 1994; Pusey and Wolf 1996; Goossens et al. 2001; Takahata et al. 2002; Archie et al. 2007; Costello et al. 2008). There is some evidence for all three mechanisms in the study population. First, most males on Cayo Santiago disperse from their natal group, leaving all familiar female maternal relatives behind (Berard 1990). Since male reproduction is skewed (Widdig et al. 2004; Dubuc et al. 2011, 2014), only a small proportion of males each produce several offspring who are then related as paternal half-siblings. Male natal dispersal at puberty should most likely separate paternal half-siblings of the opposite sex born in the same group before they start reproducing (Kuester et al. 1994); however, a delay in natal dispersal increases the chance of mating between paternal half-siblings of the opposite sex (Alberts 1999). As male rhesus macaques change groups several times during their lives (Berard et al. 1994) and reproduce in different groups, paternal half-siblings can also be born and grow up in different groups. This could potentially result in matings between unfamiliar paternal half-siblings of the opposite sex either via extra-group paternity or if males end up in a non-natal group containing paternal half-sisters, as found in

**Table 4**Results of the GLMM and LMMs testing inbreeding over time assessed by F and inbreeding estimates (IR, SH, and HL) using the samesubset of 609 individuals (four complete generation pedigrees)

	Est	SE	$\chi^2$	df	P values	2.5% CL	97.5% CL
F							
Intercept	-5.762	0.496				-11.021	-0.815
Birth season	-0.180	0.363	0.325	1	0.569	-6.563	0.508
IR							
Intercept	-0.001	0.005				-0.010	0.008
Birth season	-0.005	0.005	0.907	1	0.341	-0.014	0.003
SH							
Intercept	0.000	0.006				-0.011	0.012
Birth season	0.007	0.005	1.501	1	0.221	-0.002	0.019
HL							
Intercept	0.000	0.004				-0.008	0.007
Birth season	-0.005	0.004	1.215	1	0.270	-0.013	0.003

**Table 5**Distribution of parental relatedness in the actual breeders(based on 966 offspring)

R	0	0.0625	0.125	0.25	0.5
Ν	893	42	25	5	1

N represents the number of actual breeders per degree of relatedness (r)

this study. Father-daughter inbreeding is less expected given that sires tend to disperse before their daughters mature (Clutton-Brock 1988; Alberts and Altmann 1995). This seems also true for the study population given that mean male tenure is about 2 years (Manson 1995) and that we observed only two cases of inbreeding in father-daughter pairs. Hence, our results are in line with previous suggestions that even when only few other groups are available, dispersal should result in low inbreeding coefficients (Cheney and Seyfarth 1983).

Second, inbreeding avoidance might also be facilitated by the occurrence of extra-group paternities. Extra-group paternity could allow individuals to choose unrelated breeding partners, particularly if inbreeding risk is high in the natal group. Extra-group paternities indeed are regularly observed on Cayo Santiago, with 24% of offspring from six birth cohorts within one social group being sired by non-group males (Widdig et al. 2004). However, comprehensive studies investigating whether rates of extra-group paternity correlate with the risk of inbreeding in the natal group are lacking.

Finally, the recognition of related individuals and a preference for unrelated mating partners were proposed to aid in



parental relatedness

Fig. 2 Simulated *r*-value distributions considering male reproductive skew and extra-group paternity. While the distribution depicted by *gray bars* was simulated including natal breeders, they were excluded in the second simulation (*hatched bars*). The *arrow* indicates the observed mean parental relatedness. Comparing both distributions, this illustrates that individuals seem to avoid breeding with kin which decreases the chance of inbreeding

**Table 6**Results of the Cox regressions testing the influence of F andinbreeding estimates (IR, SH, and HL), respectively, on survival up to1 year of life

Survival within the first year of life						
Predictor	Estimate	SE	Ζ	Р		
F (4 gen)	4.954	7.065	0.700	0.480		
IR	1.259	1.417	0.890	0.370		
SH	-1.333	1.329	-1.000	0.320		
HL	1.822	1.888	0.970	0.330		

inbreeding avoidance (Widdig 2007). Evidence for kin bias (implying kin recognition) has been found for both maternal and paternal kin in the study population (Kapsalis and Berman 1996; Widdig et al. 2001). The effect of paternal kin bias was always weaker than that of maternal kin bias (reviewed in Widdig 2013), indicating that the degree of familiarity, and hence the reliability of kin recognition, is lower for paternal kin (see below). Nevertheless, there is growing evidence for kin recognition among paternal kin requiring alternative mechanisms, such as phenotype matching (Pfefferle et al. 2014a, b; Levréro et al. 2015).

Although our study lacks mating data, previous studies on mating behavior revealed that female rhesus macaques show sexual aversion to male kin and prefer to mate with unrelated males (Manson and Perry 1993). Our genetic data on reproductive output, together with our simulation, now suggest that mating between kin might actually be avoided. It is important to note that our study is unable to distinguish between precopulatory (avoidance of mating with relatives) and postcopulatory (e.g., failure to fertilize, spontaneous abortion due to mating with relatives) mechanisms. For example, father-daughter matings seem actively avoided in whitefaced capuchin monkeys, Cebus capucinus, although alpha males in this species monopolize most paternities and are often still present when daughters matured (Muniz et al. 2006). Future studies should investigate the underlying mechanism in more detail and additionally explore the possibility of early fetal loss when close paternal kin breed (cf. Muniz et al. 2006).

The results of our study revealed that 0.79% of the 2669 individuals with complete three generation pedigrees had an

**Table 7** Results of the Cox regressions testing the influence of F andinbreeding estimates (IR, SH, and HL), respectively, on survival up tosexual maturation

Predictor	Estimate	SE	Ζ	P values
F (4 gen)	2.978	5.937	0.500	0.620
IR	0.547	1.120	0.490	0.620
SH	-0.570	1.051	-0.540	0.590
HL	0.815	1.490	0.550	0.580

F > 0, which increased to 7.39% when four complete generation pedigrees of 609 individuals were considered. As theoretically expected, the proportion of inbred individuals detected increased (Balloux et al. 2004), while the average F decreased with the number of generations included, because additional inbreeding events appeared further up in the pedigree. When considering three generation pedigrees, inbreeding among paternal half-siblings was predominant, while when considering four generations (and hence inbreeding events deeper in the pedigree), the parent pair producing the inbred offspring was more often related via both the maternal and the paternal lines. Inbreeding at the half-sibling level (r = 0.25) was more likely within the paternal line. This is likely for at least two reasons. First, there is a higher availability of paternal than maternal half-siblings in general, but in particular within the same age cohort, due to male skew in annual reproduction and females mainly giving birth to one offspring per year (Widdig et al. 2004). Hence, sons of successful sires face a high number of same-aged paternal halfsisters that they can potentially mate with, particularly when dispersal is delayed. Second, paternal half-siblings grow up with different mothers (and hence different social environments), which results in lower familiarity among them compared to maternal half-siblings sharing the same mother and social setting (Widdig 2013). Similar, in most inbreeding cases between parents related via the maternal and paternal line, females mated with an unfamiliar male relative, as offspring of their male kin, even of a maternal brother, are unlikely to be familiar to them.

What seems most critical when assessing inbreeding in a given population is to evaluate potential fitness consequences due to inbreeding depression (Keller and Waller 2002). For example, in a natural population of gray seals (Halichoerus grypus), pups with a higher IR had a significantly lower survival (Bean et al. 2004). Likewise, a study of an island population of red deer (Cervus elaphus) found evidence of inbreeding depression on offspring birth weight and first year survival (Walling et al. 2011; Huisman et al. 2016). Inbreeding depression was also reported in several traits among cooperatively breeding meerkats, Suricata suricatta (Nielsen et al. 2012). Interestingly, pup survival did not show evidence of inbreeding depression, but survival of inbred juveniles was reduced, possibly due to the increased stress associated with reaching independence (Nielsen et al. 2012), which is similar to results reported for bush dogs, Speothos venaticus (Bilski et al. 2013).

Both our marker-based estimates and our inbreeding coefficient based on pedigrees detected no evidence for costs of inbreeding due to death in early infancy or until maturation (1 or 4 years of age) suggesting that there may be no inbreeding depression in the sample investigated. This is interesting given that we still detected mechanisms of inbreeding avoidance. Several possibilities are likely explanations. First, there might have been inbreeding depression in the past (potentially long before the foundation of the study population) that led to selection of inbreeding avoidance mechanisms that we find today. A second possibility is that inbreeding avoidance might not have been costly to evolve in this species and therefore may have been selected for even though costs of inbreeding are rather low, too (cf. Archie et al. 2007). Alternatively, we might have been unable to detect inbreeding depression due to a low power given the restricted number of complete generations of pedigrees or markers used (cf. Overall et al. 2005). Furthermore, food provisioning in the study population might buffer effects of inbreeding depression, despite the limited access to artificial food by infants below 1 year of age. Future studies should therefore investigate more life history stages using more comprehensive genetic data under wild conditions.

Although our results indicate that colony management did not produce a bias through selective culling of inbred individuals, our genetic sample included only 75.5% of offspring born during our entire study period (see Supplement). Hence, it might be argued that our sample could lack individuals that died due to inbreeding depression before genetic sampling. We compensated this in two ways, first by enhancing sampling effort to 87.4% of offspring when investigating the impact of inbreeding on mortality and second by restricting our analysis to individuals with complete pedigrees (except in the mortality analysis, caused by too few inbred cases). In addition, mean observed inbreeding did not differ between years with high infant sampling success (>90%) and years of lower sampling success (<90%), increasing our confidence that the lack of samples did not bias our results.

Overall, mechanisms of inbreeding avoidance appear to be effectively used within the study population, which, in combination with a sufficiently large and presumably unrelated founder population, seem to reduce inbreeding as compared to estimates from biologically relevant simulations. Our simulation further supports the importance of sex-biased dispersal as a mechanism of inbreeding avoidance as it revealed a lower expected relatedness of potential parents when excluding natal breeding. The avoidance to breed with kin, in addition to male dispersal, even with the limited options of an isolated population, seems to be sufficient to maintain heterozygosity. These results are consistent with genetic data from wild rhesus macaques reporting no inbreeding, but high gene flow between groups, likely caused by male natal dispersal, relatively short non-natal group tenure, and avoidance of consanguineous matings (Melnick et al. 1984). Both approaches, pedigree data and genetic markers, revealed the same patterns with regard to inbreeding and its potential costs. Hence, the complementary use of both approaches might lead to more robust conclusions than either method alone. This calls for more long-term studies including extended genetic, mating, and fitness data from wild or free-ranging populations as they have a great potential

to increase our understanding on inbreeding and its consequences across different social systems.

In contrast to our study, previous studies on other isolated populations with much smaller founder size showed a higher degree of inbreeding events (reviewed in Keller and Waller 2002). For instance, a pedigree study of an isolated population of bighorn sheep (Ovis canadensis) maintained at about 30 females revealed that 18.1% of the lambs had an F > 0(Rioux-Paquette et al. 2011). In addition, the probability of lamb survival for inbred females was 40% lower than for non-inbred ones (Rioux-Paquette et al. 2011). The only extensive pedigree data of a primate species so far come from an isolated mandrill (Mandrillus sphinx) group, with five complete generations of pedigrees including 14 founders (Charpentier et al. 2006). In this population, 30% of all offspring born were inbred, with F ranging from 0.0625 to 0.25 (Charpentier et al. 2006). These higher levels of inbreeding were probably caused by a more pronounced male reproductive skew in mandrills (Charpentier et al. 2005a), by the lack of dispersal in this population, and potentially also by the limited number of founders. Our results might therefore be helpful to determine an appropriate size of a given founder population in order to reduce the risk of inbreeding. However, effects of inbreeding cannot be compared directly between species, as many factors such as genetic diversity, dispersal and mating patterns, kin structure, and environmental conditions influence the potential of inbreeding (Hedrick and Kalinowski 2000; Keller et al. 2002; Edmands 2007; Pemberton 2008). Consequently, it is difficult to give a general number for the minimum or optimal founder population size, although such estimates would be desired for managed or fragmented populations. Given that many animal species face dramatic habitat loss combined with critical population declines (e.g., Campbell et al. 2008) and a loss of connectivity between subpopulations, this would be particularly important for primates serving as umbrella species.

Acknowledgements Open access funding provided by Max Planck Society. We thank Edgar Davila, Julio Resto, and Giselle Caraballo Cruz for assistance during data collection, Elizabeth Maldonado for data management, and Cecilia Penedo for generating the CPRC genetic database. The genetic consortium including CPRC (Fred Bercovitch, Matt Kessler, John Berard) and German scientists (Michael Krawczak, Peter Nürnberg, and Jörg Schmidtke) is acknowledged for their effort to start the Cayo Santiago genetic database in the 1990s. Furthermore, we are grateful to Andrea Trefilov, Elisabeth Kirst, Tanja Heinz, and Sarah Nagel for improving the UL genetic database and in particular to Linda Vigilant for laboratory access and input into this analysis. Brigitte Weiß, Elise Huchard, and two anonymous reviewers provided fruitful comments on an earlier version of the manuscript. The study was conducted within the Jr. Research Group of Primate Kin Selection, an Emmy-Noether Group funded by the German Research Foundation (DFG, grant Wi 1808/1-1, 1-2, 2-1, 3-1, 5-1 awarded to AW). We thank the Max-Planck Institute for Evolutionary Anthropology and the University of Leipzig for logistic support. The rhesus population is currently supported by the National Center for Research Resources (NCRR, grant 8P40OD012217), the Office of Research Infrastructure Programs (ORIP) of the National Institutes of Health. The content of this publication is solely in the responsibility of the authors and does not necessarily represent the official views of NCRR or ORIP.

#### Compliance with ethical standards

**Funding** This research was supported by the German Research Foundation (DFG), within the Emmy-Noether Program (grant Wi 1808/1-1, 1-2, 2-1, 3-1, 5-1 awarded to AW). The population of Cayo Santiago is currently supported by the National Center for Research Resources (NCRR, grant 8P40OD012217), the Office of Research Infrastructure Programs (ORIP) of the National Institutes of Health. The content of this publication is solely the responsibility of the authors and does not necessarily represent the official views of NCRR or ORIP.

**Conflict of interest** The authors declare that they have no conflict of interest.

**Ethical statement** All applicable international, national, and/or institutional guidelines for the care and use of animals were followed. All research procedures of this study were approved by the CPRC and the Institutional Animal Care and Use Committee (IACUC) of the University of Puerto Rico (protocol number 4060105). The sample transfer was approved as required (Cites export permissions no. 05US101361/9, no. 06US112079/9, no. 07US133766/9, no. 08US163309/9, no. 09US200870/9, no. 09US20435/9, and no. 11US28371A/9 and Cites import permission no. E-1426/05, no. E-1082/06, no. E-1207/07, no. E-1215/08, no. E01146/09, no. E-00049/10, and no. E-00836/11).

**Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

# References

- Acevedo-Whitehouse K, Gulland F, Greig D, Amos W (2003) Inbreeding: disease susceptibility in California sea lions. Nature 422:35
- Adams DC, Anthony CD (1996) Using randomization techniques to analyse behavioural data. Anim Behav 51:733–738
- Alberts SC (1999) Paternal kin discrimination in wild baboons. Proc R Soc Lond B 266:1501–1506
- Alberts SC, Altmann J (1995) Balancing costs and opportunities: dispersal in male baboons. Am Nat 145:279–306
- Alho JS, Välimäki K, Merilä J (2010) Rhh: an R extension for estimating multilocus heterozygosity and heterozygosity–heterozygosity correlation. Mol Ecol Resour 10:720–722
- Altmann SA (1962) A field study of the sociobiology of rhesus monkeys, Macaca mulatta. Ann N Y Acad Sci 102:338–435
- Amos W, Worthington Wilmer J, Fullard K, Burg TM, Croxall JP, Bloch D, Coulson T (2001) The influence of parental relatedness on reproductive success. Proc R Soc Lond B 268:2021–2027
- Aparicio JM, Ortego J, Cordero PJ (2006) What should we weigh to estimate heterozygosity, alleles or loci? Mol Ecol 15:4659–4665
- Archie EA, Hollister-Smith JA, Poole JH, Lee PC, Moss CJ, Maldonado JE, Fleischer RC, Alberts SC (2007) Behavioural inbreeding avoidance in wild African elephants. Mol Ecol 16:4138–4148

- Baayen H (2008) Analyzing linguistic data: a practical introduction to statistics using R. Cambridge University Press, Cambridge
- Balloux F, Amos W, Coulson T (2004) Does heterozygosity estimate inbreeding in real populations? Mol Ecol 13:3021–3031
- Barr DJ, Levy R, Scheepers C, Tily HJ (2013) Random effects structure for confirmatory hypothesis testing: keep it maximal. J Mem Lang 68:255–278
- Bates D, Maechler M, Bolker B, Walker S (2015) lme4: linear mixedeffects models using Eigen and S4. R package version 1:0–8 http:// CRAN.R-project.org/package=lme4
- Bean K, Amos W, Pomeroy PP, Twiss SD, Coulson TN, Boyd IL (2004) Patterns of parental relatedness and pup survival in the grey seal (*Halichoerus grypus*). Mol Ecol 13:2365–2370
- Bensch S, Andrén H, Hansson B, Pedersen HC, Sand H, Sejberg D, Wabakken P, Åkesson M, Liberg O (2006) Selection for heterozygosity gives hope to a wild population of inbred wolves. PLoS One 1:e72
- Berard JD (1990) Life history patterns of male rhesus macaques on Cayo Santiago. PhD dissertation. University of Oregon, Oregon
- Berard JD, Nurnberg P, Epplen JT, Schmidtke J (1994) Alternative reproductive tactics and reproductive success in male rhesus macaques. Behaviour 129:177–201
- Bilski DR, Pie MR, Passos FC (2013) Variable inbreeding effects across life-history stages in a captive carnivorous mammal population. Anim Conserv 16:633–640
- Blomquist GE (2013) Maternal effects on offspring mortality in rhesus macaques (Macaca mulatta). Am J Primatol 75:238–251
- Campbell G, Kuehl H, Kouame PN, Boesch C (2008) Alarming decline of West African chimpanzees in Cote d'Ivoire. Curr Biol 18:R903– R904
- Carpenter CR (1972) Breeding colonies of macaques and gibbons on Santiago Island, Puerto Rico. In: Beveredge W (ed) Breeding primates. Karger, Basel, pp. 76–87
- Carpenter CR, Krakower CA (1941) Notes on results of a test for tuberculosis in rhesus monkeys (*Macaca mulatta*). PR J Public Health Trop Med 17:3–13
- Cassinello J (2005) Inbreeding depression on reproductive performance and survival in captive gazelles of great conservation value. Biol Conserv 122:453–464
- Charlesworth D, Charlesworth B (1987) Inbreeding depression and its evolutionary consequences. Annu Rev Ecol Syst 18:237–268
- Charpentier M, Peignot P, Hossaert-Mckey M, Gimenez O, Setchell JM, Wickings EJ (2005a) Constraints on control: factors influencing reproductive success in male mandrills. Behav Ecol 16:614–623
- Charpentier M, Setchell JM, Prugnolle F, Knapp LA, Wickings EJ, Peignot P, Hossaert-Mckey M (2005b) Genetic diversity and reproductive success in mandrills (*Mandrillus sphinx*). P Natl Acad Sci USA 102:16723–16728
- Charpentier M, Setchell JM, Prugnolle F, Wickings J, Peignot P, Balloux F, Hossaert-Mckey M (2006) Life history correlates of inbreeding depression in mandrills (*Mandrillus sphinx*). Mol Ecol 15:21–28
- Cheney DL, Seyfarth RM (1983) Nonrandom dispersal in free-ranging vervet monkeys: social and genetic consequences. Am Nat 122: 392–412
- Clutton-Brock TH (1988) Reproductive success: studies of individual variation in contrasting breeding systems. University of Chicago Press, Chicago
- Coltman DW, Pilkington JG, Smith JA, Pemberton JM (1999) Parasitemediated selection against inbred Soay sheep in a free-living, island population. Evolution 53:1259–1267
- Colvin J (1983) Influences of the social situation on male emigration. In: Hinde RA (ed) Primate social relationship. An integrated approach. Blackwell, Oxford, pp. 160–171
- Costello CM, Creel SR, Kalinowski ST, Vu NV, Quigley HB (2008) Sexbiased natal dispersal and inbreeding avoidance in American black

bears as revealed by spatial genetic analyses. Mol Ecol 17:4713-4723

- Coster A (2012) pedigree: pedigree functions. R package version 1.4, http://CRAN.R-project.org/package=pedigree
- Dixson AF, Nevison CM (1997) The socioendocrinology of adolescent development in male rhesus monkeys (*Macaca mulatta*). Horm Behav 31:126–135
- Dorostkar M, Faraji Arough H, Shodja J, Rafat SA, Rokouei M, Esfandyari H (2012) Inbreeding and inbreeding depression in Iranian Moghani sheep breed. J Agric Sci Technol 14:549–556
- Dubuc C, Muniz L, Heistermann M, Engelhardt A, Widdig A (2011) Testing the priority-of-access model in a seasonally breeding primate species. Behav Ecol Sociobiol 65:1615–1627
- Dubuc C, Ruiz-Lambides A, Widdig A (2014) Variance in male lifetime reproductive success and estimation of the degree of polygyny in a primate. Behav Ecol 25:878–889
- Duggleby CR, Haseley PA, Rawlins RG, Kessler MJ (1986) An overview of blood group genetic studies on the Cayo Santiago rhesus monkeys. In: Rawlins RG, Kessler MJ (eds) The Cayo Santiago macaques: history, behavior, and biology. State University of New York Press, Albany, pp. 269–282
- Edmands S (2007) Between a rock and a hard place: evaluating the relative risks of inbreeding and outbreeding for conservation and management. Mol Ecol 16:463–475
- Forstmeier W, Schielzeth H, Mueller JC, Ellegren H, Kempenaers B (2012) Heterozygosity–fitness correlations in zebra finches: microsatellite markers can be better than their reputation. Mol Ecol 21: 3237–3249
- Fredrickson R, Hedrick P (2002) Body size in endangered Mexican wolves: effects of inbreeding and cross-lineage matings. Anim Conserv 5:39–43
- Goossens B, Chikhi L, Taberlet P, Waits LP, Allaine D (2001) Microsatellite analysis of genetic variation among and within Alpine marmot populations in the French Alps. Mol Ecol 10:41–52
- Gouzoules S, Gouzoules H (1987) Kinship. In: Smuts BB, Cheney DL, Seyfarth RM, Wrangham RW, Struhsaker TT (eds) Primate societies. University of Chicago Press, Chicago, pp. 299–305
- Grueber CE, Laws RJ, Nakagawa S, Jamieson IG (2010) Inbreeding depression accumulation across life-history stages of the endangered Takahe. Conserv Biol 24:1617–1625
- Hedrick PW, Kalinowski ST (2000) Inbreeding depression in conservation biology. Annu Rev Ecol Syst 31:139–162
- Hernández-Pacheco R, Rawlins RG, Kessler MJ, Williams LE, Ruiz-Maldonado TM, González-Martínez J, Ruiz-Lambides AV, Sabat AM (2013) Demographic variability and density-dependent dynamics of a free-ranging rhesus macaque population. Am J Primatol 75: 1152–1164
- Hoffman CL, Ruiz-Lambides AV, Davila E, Maldonado E, Gerald MS, Maestripieri D (2008) Sex differences in survival costs of reproduction in a promiscuous primate. Behav Ecol Sociobiol 62:1711–1718
- Hoffman JI, Simpson F, David P, Rijks JM, Kuiken T, Thorne MAS, Lacy RC, Dasmahapatra KK (2014) High-throughput sequencing reveals inbreeding depression in a natural population. P Natl Acad Sci USA 111:3775–3780
- Huisman J, Kruuk LEB, Ellis PA, Clutton-Brock T, Pemberton JM (2016) Inbreeding depression across the lifespan in a wild mammal population. P Natl Acad Sci USA 113:3585–3590
- Kalinowski ST, Taper ML, Marshall TC (2007) Revising how the computer program CERVUS accommodates genotyping error increases success in paternity assignment. Mol Ecol 16:1099–1106
- Kanthaswamy S, Ng J, Penedo MCT, Ward T, Smith DG, Ha JC (2012) Population genetics of the Washington National Primate Research Center's (WaNPRC) captive pigtailed macaque (*Macaca nemestrina*) population. Am J Primatol 74:1017–1027

- Kapsalis E, Berman CM (1996) Models of affiliative relationships among free-ranging rhesus monkeys (*Macaca mulatta*) I. Criteria for kinship Behaviour 133:1209–1234
- Kardos M, Luikart G, Allendorf FW (2015) Measuring individual inbreeding in the age of genomics: marker-based measures are better than pedigrees. Heredity 115:63–72
- Keller LF, Arcese P (1998) No evidence for inbreeding avoidance in a natural population of song sparrows (*Melospiza melodia*). Am Nat 152:380–392
- Keller LF, Grant PR, Grant BR, Petren K (2002) Environmental conditions affect the magnitude of inbreeding depression in survival of Darwin's finches. Evolution 56:1229–1239
- Keller LF, Waller DM (2002) Inbreeding effects in wild populations. Trends Ecol Evol 17:230–241
- Kim SH, Cheng KM-T, Ritland C, Ritland K, Silversides FG (2007) Inbreeding in Japanese quail estimated by pedigree and microsatellite analyses. J Hered 98:378–381
- Knief U, Hemmrich-Stanisak G, Wittig M, Franke A, Griffith SC, Kempenaers B, Forstmeier W (2015) Quantifying realized inbreeding in wild and captive animal populations. Heredity 114:397–403
- Kuester J, Paul A, Arnemann J (1994) Kinship, familiarity and mating avoidance in Barbary macaques (*Macaca sylvanus*). Anim Behav 48:1183–1194
- Levréro F, Carrete-Vega G, Herbert A, Lawabi I, Courtiol A, Willaume E, Kappeler PM, Charpentier MJE (2015) Social shaping of voices does not impair phenotype matching of kinship in mandrills. Nat Commun 6:7609
- Li DY, Xu HL, Trask JS, Zhu Q, Cheng AC, Smith DG, George D, Zhang L (2013) Genetic diversity and population structure in wild Sichuan rhesus macaques. Mol Biol Rep 40:3033–3041
- Liberg O, Andren H, Pedersen H, Sand H, Sejberg D, Wabakken P, Akesson M, Bensch S (2005) Severe inbreeding depression in a wild wolf (*Canis lupus*) population. Biol Lett 1:17–20
- Lindburg DG (1969) Rhesus monkeys: mating season mobility of adult males. Science 166:1176–1178
- Manly BFJ (1997) Randomization, bootstrap and Monte Carlo methods in biology. Chapman & Hall, London
- Manson JH (1995) Do female rhesus macaques choose novel males? Am J Primatol 37:285–296
- Manson J, Perry SE (1993) Inbreeding avoidance in rhesus macaques: whose choice? Am J Primatol 90:335–344
- Marriott BM, Roemer J, Sultana C (1989) An overview of the food intake patterns of the Cayo Santiago rhesus monkeys (*Macaca mulatta*): report of a pilot study. P R Health Sci J 8:87–94
- Marshall TC, Coltman DW, Pemberton JM, Slate J, Spalton JA, Guinness FE, Smith JA, Pilkington JG, Clutton-Brock TH (2002) Estimating the prevalence of inbreeding from incomplete pedigrees. Proc R Soc Lond B 269:1533–1539
- Marshall TC, Spalton JA (2000) Simultaneous inbreeding and outbreeding depression in reintroduced Arabian oryx. Anim Conserv 3:241– 248
- Melnick DJ, Pearl MC, Richard AF (1984) Male migration and inbreeding avoidance in wild rhesus monkeys. Am J Primatol 7:229–243
- Muniz L, Perry S, Manson JH, Gilkenson H, Gros-Louis J, Vigilant L (2006) Father-daughter inbreeding avoidance in a wild primate population. Curr Biol 16:R156–R157
- Nakagawa S, Schielzeth H (2013) A general and simple method for obtaining R2 from generalized linear mixed-effects models. Methods Ecol Evol 4:133–142
- Nielsen JF, English S, Goodall-Copestake WP et al (2012) Inbreeding and inbreeding depression of early life traits in a cooperative mammal. Mol Ecol 21:2788–2804
- Ólafsdóttir G, Kristjánsson T (2008) Correlated pedigree and molecular estimates of inbreeding and their ability to detect inbreeding depression in the Icelandic sheepdog, a recently bottlenecked population of domestic dogs. Conserv Genet 9:1639–1641

- Pemberton J (2004) Measuring inbreeding depression in the wild: the old ways are the best. Trends Ecol Evol 19:613–615
- Pemberton JM (2008) Wild pedigrees: the way forward. Proc R Soc Lond B 275:613–621
- Pfefferle D, Kazem AJN, Brockhausen RR, Ruiz-Lambides AV, Widdig A (2014a) Monkeys spontaneously discriminate their unfamiliar paternal kin under natural conditions using facial cues. Curr Biol 24:1806–1810
- Pfefferle D, Ruiz-Lambides AV, Widdig A (2014b) Female rhesus macaques discriminate unfamiliar paternal sisters in playback experiments—support for acoustic phenotype matching. Proc R Soc B 281:20131628
- Powell JE, Visscher PM, Goddard ME (2010) Reconciling the analysis of IBD and IBS in complex trait studies. Nat Rev Genet 11:800-805
- Prado-Martinez J, Sudmant PH, Kidd JM et al (2013) Great ape genetic diversity and population history. Nature 499:471–475
- Pusey AE, Wolf M (1996) Inbreeding avoidance in animals. Trends Ecol Evol 11:201–206
- Core Team R (2015) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna R package version 3.2.1, http://www.R-project.org/
- Development Core Team R (2010) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna R package version 2.15.1, http://www.R-project.org/
- Ralls K, Ballou JD, Templeton A (1988) Estimates of lethal equivalents and the cost of inbreeding in mammals. Conserv Biol 2: 185–193
- Rawlins RG, Kessler MJ (eds) (1986) The Cayo Santiago macaques. History, behavior and biology. State University of New York Press, Albany
- Rijks JM, Hoffman JI, Kuiken T, Osterhaus A, Amos W (2008) Heterozygosity and lungworm burden in harbour seals (*Phoca vitulina*). Heredity 100:587–593
- Rioux-Paquette E, Festa-Bianchet M, Coltman DW (2010) No inbreeding avoidance in an isolated population of bighorn sheep. Anim Behav 80:865–871
- Rioux-Paquette E, Festa-Bianchet M, Coltman DW (2011) Sexdifferential effects of inbreeding on overwinter survival, birth date and mass of bighorn lambs. J Evol Biol 24:121–131
- Ruiz-López MJ, Roldán ERS, Espeso G, Gomendio M (2009) Pedigrees and microsatellites among endangered ungulates: what do they tell us? Mol Ecol 18:1352–1364
- Rzewuska K, Klewiec J, Martyniuk E (2005) Effect of inbred on reproduction and body weight of sheep in a closed Booroola flock. Anim Sci Pap Rep 23:237–247
- Satkoski J, George D, Smith DG, Kanthaswamy S (2008) Genetic characterization of wild and captive rhesus macaques in China. J Med Primatol 37:67–80
- Schielzeth H, Forstmeier W (2009) Conclusions beyond support: overconfident estimates in mixed models. Behav Ecol 20: 416-420
- Seth PK, Seth S (1986) Ecology and behavior of rhesus monkeys in India. In: Else JG, Lee PC (eds) Primate ecology and conservation. Cambridge University Press, Cambridge, pp. 192–305
- Slate J, David P, Dodds K, Veenvliet B, Glass B, Broad T, McEwan J (2004) Understanding the relationship between the inbreeding coefficient and multilocus heterozygosity: theoretical expectations and empirical data. Heredity 93:255–265
- Slate J, Kruuk LEB, Marshall TC, Pemberton JM, Clutton-Brock TH (2000) Inbreeding depression influences lifetime breeding success

in a wild population of red deer (*Cervus elaphus*). Proc R Soc Lond B 267:1657–1662

- Slate J, Pemberton JM (2002) Comparing molecular measures for detecting inbreeding depression. J Evol Biol 15:20–31
- Sugiyama Y, Kurita H, Matsui T, Kimoto S, Shimomura T (2009) Carrying of dead infants by Japanese macaque (*Macaca fuscata*) mothers. Anthropol Sci 117:113–119
- Szulkin M, Garant D, Mccleery RH, Sheldon BC (2007) Inbreeding depression along a life-history continuum in the great tit. J Evol Biol 20:1531–1543
- Szulkin M, Zelazowski P, Nicholson G, Sheldon BC (2009) Inbreeding avoidance under different null models of random mating in the great tit. J Anim Ecol 78:778–788
- Takahata Y, Huffman MA, Bardi M (2002) Long-term trends in matrilineal inbreeding among the Japanese macaques of Arashiyama B troop. Int J Primatol 23:399–410
- Taylor HR, Kardos MD, Ramstad KM, Allendorf FW (2015) Valid estimates of individual inbreeding coefficients from marker-based pedigrees are not feasible in wild populations with low allelic diversity. Conserv Genet 16:901–913
- Taylor SS, Sardell RJ, Reid JM, Bucher T, Taylor HG, Arcese P, Keller LF (2010) Inbreeding coefficient and heterozygosity–fitness correlations in unhatched and hatched song sparrow nestmates. Mol Ecol 19:4454–4461
- Therneau TM (2015) coxme: mixed effects Cox models. R package version 2.2–5, http://CRAN.R-project.org/package=coxme

- Walling CA, Nussey DH, Morris A, Clutton-Brock TH, Kruuk LE, Pemberton JM (2011) Inbreeding depression in red deer calves. BMC Evol Biol 11:318
- Wang J (2011) A new likelihood estimator and its comparison with moment estimators of individual genome-wide diversity. Heredity 107: 433–443
- Widdig A (2007) Paternal kin discrimination: the evidence and likely mechanisms. Biol Rev 82:319–334
- Widdig A (2013) The impact of male reproductive skew on kin structure and sociality in multi-male groups. Evol Anthropol 22:239–250
- Widdig A, Bercovitch FB, Streich WJ, Sauermann U, Nürnberg P, Krawczak M (2004) A longitudinal analysis of reproductive skew in male rhesus macaques. Proc R Soc Lond B 271: 819–826
- Widdig A, Kessler MJ, Bercovitch FB et al (2016) Genetic studies on the Cayo Santiago rhesus macaques: a review of 40 years of research. Am J Primatol 78:44–62
- Widdig A, Nürnberg P, Krawczak M, Streich WJ, Bercovitch FB (2001) Paternal relatedness and age proximity regulate social relationships among adult female rhesus macaques. P Natl Acad Sci USA 98: 13769–13773
- Zehr JL, Meter PEV, Wallen K (2005) Factors regulating the timing of puberty onset in female rhesus monkeys (*Macaca mulatta*): role of prenatal androgens, social rank, and adolescent body weight. Biol Reprod 72:1087–1094